Author's response to reviews

Title: Management of Giant Pseudomeningoceles after Spinal Surgery

Authors:

Yi-Jan Weng (wengyj@adm.cgmh.org.tw)
Chin-Chang Cheng (chengcc@adm.cgmh.org.tw)
Yen-Yao Li (orthoyao@adm.cgmh.org.tw)
Tsung-Jen Huang (tjhuang@adm.cgmh.org.tw)
Robert Wen-Wei Hsu (wwh@adm.cgmh.org.tw)

Version: 2 Date: 27 June 2009

Author's response to reviews: see over
Dear Editor-in-Chief,

Enclosed is our article entitled “Management of Giant Pseudomeningoceles after Spinal Surgery” which we are submitting for publication in The Journal.

In consideration of *BMC Musculoskeletal Disorders*, reviewing and editing our submission, the authors undersigned hereby transfer, assign or otherwise convey all copyright ownership to *BMC Musculoskeletal Disorders*, and represent that they own all rights in material submitted. The authors further represent that the article is original, that it is not under consideration by another journal and that it has not been previously published. This assignment is to take effect only in the event that such work is published in *The Journal*. Each of the authors represents that he has read and approved that final manuscript.

Sincerely,

Tsung-Jen Huang, MD